

Citation: Zhao S, Stone L, Gao D, He D (2018) Modelling the large-scale yellow fever outbreak in Luanda, Angola, and the impact of vaccination. PLoS Negl Trop Dis 12(1): e0006158. <u>https://doi.</u> org/10.1371/journal.pntd.0006158

Editor: Isabel Rodriguez-Barraquer, University of California San Francisco, UNITED STATES

Received: June 20, 2017

Accepted: December 11, 2017

Published: January 16, 2018

Copyright: © 2018 Zhao et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: DH was supported by Start-up grant for new recruits of Hong Kong Polytechnic University and the Early Career Scheme from Hong Kong Research Grants Council (PolyU 251001/14M). DG was supported by NSFC (11601336), Program for Professor of Special Appointment (Eastern Scholar) at Shanghai Institutions of Higher Learning (TP2015050), Shanghai Gaofeng Project for University Academic Development Program. RESEARCH ARTICLE

Modelling the large-scale yellow fever outbreak in Luanda, Angola, and the impact of vaccination

Shi Zhao¹, Lewi Stone^{2,3}*, Daozhou Gao⁴, Daihai He¹*

1 Department of Applied Mathematics, Hong Kong Polytechnic University, Hong Kong, China, 2 School of Mathematical and Geospatial Sciences, RMIT University, Melbourne, Australia, 3 Biomathematics Unit, Department of Zoology, Tel Aviv University, Ramat Aviv, Israel, 4 Department of Mathematics, Shanghai Normal University, Shanghai, China

* lewi.stone@rmit.edu.au (LS); daihai.he@polyu.edu.hk (DH)

Abstract

Background

Yellow fever (YF), transmitted via bites of infected mosquitoes, is a life-threatening viral disease endemic to tropical and subtropical regions of Africa and South America. YF has largely been controlled by widespread national vaccination campaigns. Nevertheless, between December 2015 and August 2016, YF resurged in Angola, quickly spread and became the largest YF outbreak for the last 30 years. Recently, YF resurged again in Brazil (December 2016). Thus, there is an urgent need to gain better understanding of the transmission pattern of YF.

Model

The present study provides a refined mathematical model, combined with modern likelihood-based statistical inference techniques, to assess and reconstruct important epidemiological processes underlying Angola's YF outbreak. This includes the outbreak's attack rate, the reproduction number (\mathcal{R}_0), the role of the mosquito vector, the influence of climatic factors, and the unusual but noticeable appearance of two-waves in the YF outbreak. The model explores actual and hypothetical vaccination strategies, and the impacts of possible human reactive behaviors (e.g., response to media precautions).

Findings

While there were 73 deaths reported over the study period, the model indicates that the vaccination campaign saved 5.1-fold more people from death and saved from illness 5.6-fold of the observed 941 cases. Delaying the availability of the vaccines further would have greatly worsened the epidemic in terms of increased cases and deaths. The analysis estimated a mean $\mathcal{R}_0 \approx 2.6$ -3.4 and an attack rate of 0.09-0.15% (proportion of population infected) over the whole period from December 2015 to August 2016. Our estimated lower and upper bounds of \mathcal{R}_0 are in line with previous studies. Unusually, \mathcal{R}_0 oscillated in a manner that was "delayed" with the reported deaths. High recent number of deaths were associated (followed) with periods of relatively low disease transmission and low \mathcal{R}_0 , and vice-versa. The

The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

time-series of Luanda's YF cases suggest the outbreak occurred in two waves, a feature that would have become far more prominent had there been no mass vaccination. The waves could possibly be due to protective reactive behavioral changes of the population affecting the mosquito population. The second wave could well be an outcome of the March-April rainfall patterns in the 2016 El Niño year by creating ideal conditions for the breeding of the mosquito vectors. The modelling framework is a powerful tool for studying future YF epidemic outbreaks, and provides a basis for future vaccination campaign evaluations.

Author summary

An epidemic model for the transmission of yellow fever virus (YFV) in urban areas is formulated and implemented to study the 2016 yellow fever (YF) outbreak in Luanda, Angola. We explore the complex vector-host dynamics of this system taking into account mosquito abundance, vaccination and asymptomatic infections in the human population, that are generally not included in other modelling studies of YF. The model successfully fits the time series of weekly reported YF cases and deaths during the epidemic. This allows us to study the impact of the vaccination campaign in Luanda and hypothetical "delayed vaccination" scenarios. The transmission of YFV appeared to be oscillatory having a wave-like pattern in the basic reproduction number ($\mathcal{R}_0(t)$). The oscillations are hypothesized to be due to human reaction to the reported deaths, as has been noted for other human infectious diseases, and the second wave also possibly due to El Niño rainfall patterns. We conclude that the lives saved due to the vaccination campaign before August 2016 should have been approximately 370 (i.e., approximately five-fold of the observed 73 deaths), and would have been far larger extrapolating beyond August 2016.

Introduction

Yellow fever (YF) is a life-threatening viral disease endemic to tropical regions of Africa and South America. The disease is transmitted in urban areas primarily via the bites of infected female Aedes aegypti mosquitoes, which is also the vector of dengue, chikungunya and Zika viruses [1–3]. Rural and intermediate YF are transmitted by sylvatic and peri-domestic Aedes species in Africa. For those infected with YF, the disease incubates in the first 3-6 days of onset, after which there is an abrupt "period of infection" of intense viremia lasting for 3-4 days (fever, weakness, headache, nausea, muscle pain) [4]. This is followed by a period of remission in which the symptoms reduce and settle, and most infected individuals recover at this stage. Thus, some 70-85% of YF infections are asymptomatic or have at most very mild symptoms (i.e., clinically inapparent). However, 15-25% of patients relapse and move to a "period of intoxication" characterized by abdominal pain, vomiting, jaundice (yellow skin and eyes) and often culminating in death. The case-fatality-ratio (CFR) in this latter subset is understood to be approximately 20% among the general population, and 50% among hospitalized cases [4], although the CFR is well known to be highly variable, and dependent on the particular circumstances. Like Ebola, YF is classified as a viral hemorrhagic fever, although it is responsible for a 1000-fold more illness and death than Ebola [1].

In 2016, YF resurged in Angola to become the largest YF outbreak on record over the last 30 years [5]. In swift response, almost all global stocks of the YF vaccine were exhausted by

April 2016. Similar to the Angolan experience, YF recently resurged in Brazil in December 2016, where it continues to expand towards the Atlantic coast in regions not previously deemed at risk (as of March 16, 2017) [2]. Thus there is an urgent need to gain a better understanding of the transmission patterns of YF. Here we develop a mathematical model to help identify the key epidemic processes behind the Angolan outbreak in 2015-16, and the impact and effectiveness of the vaccination campaign.

The first cases of YF in Angola were seen on December 5, 2015 but reported in the media only on January 20, 2016 [6]. By November 2016, the large YF epidemic of Angola and Democratic Republic of Congo, resulted in 962 confirmed infections including 393 reported deaths [2]. YF is vaccine-preventable and the vaccine can confer long-lasting immunity. The vaccine is suitable for individuals of age 9 months or older. As such, the Angolan government initiated a vaccination campaign to prevent the spread of yellow fever on the first week of February, 2016 [2, 7]. More than 10 million doses were needed for the whole country [6]. The center of the outbreak was in Angola's capital, Luanda province. Estimates suggest that vaccination coverage of Luanda province was 38.0% at the end of January 2016, and reached 80.0% by mid-March 2016, and 93.0% by mid-June 2016 [2, 8–10].

Fig 1 graphs the epidemic curve of YF case numbers (probable and confirmed; as defined in Data section) in Luanda province as obtained from the WHO [8, 10]. The graph peaks in February 2016, when large-scale vaccination was introduced, and then followed by a period of rapid decline in case numbers. Despite the major vaccination effort, the epidemic proved tenacious rather than die out as predicted, and persisted for a sustained period of time forming a long "tail" in reported case numbers from April to August (see Fig 1). Also unusual is the minor peak in case numbers that occurred in May, followed soon after by an increase in deaths, despite the pressure of the vaccination and control efforts. By modelling and fitting YF time series of Luanda, our goal is to reconstruct the important epidemiological processes that help explain these different and sometimes nonintuitive features. The model allows estimation of the attack rate of the outbreak, and the basic reproduction number ($\mathcal{R}_0(t)$), which was changing during the epidemic. Moreover, the model is able to explore the role of the mosquito vector, and the unusual waves of the YF outbreak, which we find would have become even more apparent had there been no vaccination. Some exploration of the role of climatic variables is also possible.

As it is well known, the basic reproduction number (\mathcal{R}_0) is an important parameter to measure diseases' transmissibility, and is one of the first parameters that needs to be estimated in any epidemiological study. Recall that \mathcal{R}_0 is defined as the number of secondary cases a single typical infected individual infects over the course of its infectious period [11]. A recent study estimated \mathcal{R}_0 to lie between 5.2-7.1 at the early stage of the 2016 YF outbreak in Angola [12]. However, \mathcal{R}_0 was found to decrease with time as the epidemic proceeded. Kraemer *et al.* [13] estimated \mathcal{R}_0 to be 4.8 (95% C.I.: 4.0-5.6) for Angola, although this was possibly an over-estimate given reporting rates were not stable. In summary, the literature suggests that YF is highly transmissible with direct estimates of the reproduction number being $\mathcal{R}_0 \approx 5$, which is almost double that of pandemic influenza (\mathcal{R}_0 is from 1.5 to 3.6 [14–17]) and Ebola (\mathcal{R}_0 is from 1.2 to 2.0 [18–21]). In this work, our analysis uses modern statistical inference techniques to estimate \mathcal{R}_0 from the time-series in Fig 1. Unlike other modelling studies, our procedure also examines how reactive protective behavior (e.g., insecticide, vector-control, travel restrictions possibly in response to news and media precautions), may lead to changes in \mathcal{R}_0 , and allows us to explore the implications of this reaction.

Any model of YF must take into account that most infected individuals are asymptomatic or mild-symptomatic (individuals who show only fever but no jaundice) [3, 22–26], making the disease difficult to detect and under-reported in the first phase. With only a slight abuse of



Fig 1. The observed yellow fever outbreak in Luanda from December 5, 2015 to August 18, 2016. Green dots denote the sum of probable and confirmed cases; black dots denote confirmed cases; red bars denote the confirmed death; and blue dots denote vaccine coverage. The vertical grey dashed line denotes the time point when the YF vaccination campaign was initiated.

https://doi.org/10.1371/journal.pntd.0006158.g001

terminology, it simplifies the modelling that follows, to classify mild symptomatic individuals as asymptomatic cases. Thus asymptomatic cases refer to all individuals who do not have severe YF (i.e., without clinically apparent symptoms). It is well understood that asymptomatic YF infections can be infectious and therefore may act as "silent sources" of YFV [23]. Asymptomatic infections, thus, have the potential to play an important role in disease transmission. It was previously understood that 6 out of 7 YF infections could be asymptomatic [26]. However, a recent meta-analysis based on 11 independent studies, suggested that the asymptomatic ratio should be 55% [27]. Given the lack of information on the proportion and infectivity of asymptomatic YF cases, we examine a number of different relevant scenarios.

To the best of our knowledge, this is the first detailed modelling of YF that includes both the host and vector populations, and the asymptomatic and severe (those exhibiting fever and jaundice) cases in the host population. Previous models that assessed vaccination impact on YF have not included these fundamental components and pathways in a comprehensive approach. By fitting the time-series of the Angola outbreak, its evolution over time and its curtailment with vaccination, it becomes possible to statistically infer key model parameters. This in turn makes it possible to simulate alternative "what if" scenarios, and examine what might have happened under different vaccination schemes.

Materials and methods

Data and case definitions

We study time-series of YF cases from the province of Luanda of Angola with a population of 6,543,000 in 2016 [2, 8, 28]. The African Health Observatory (AHO) published weekly YF data for Luanda province reporting 941 (confirmed and probable) cases and 73 deaths over the study period from December 5, 2015 to August 18, 2016.

Probable cases (see [29]) are those "with acute onset of fever, with jaundice appearing within 14 days of onset of the first symptoms and one of the followings: (i) presence of yellow fever IgM antibody in the absence of YF immunization within 30 days before onset of illness; or (ii) positive postmortem liver histopathology; or (iii) epidemiological link to a confirmed case or an outbreak." Confirmed cases are defined as those positive to serological or PCR testing.

Similar to the WHO [8] and Kraemer *et al.* [13], both (weekly) probable cases and confirmed cases are grouped together and are referred to simply as "YF cases" or equivalently "severe cases" in this study. YF vaccination coverage in Luanda province, obtained from AHO reports, increased from 38% on February 2, 2016 when the vaccination campaign started to 93% on August 18, 2016 (see Fig 1) [8]. The vaccination coverage was determined by a linear interpolation of reported data (see the blue dotted line in Fig 1).

Methods

Yellow fever model. Since YFV is not spread by human-to-human-transmission, the standard SIR type modelling approaches (which are based solely on human-to-human transmission) are inappropriate. Instead we use a vector-host model of YFV transmission, as illustrated in Fig 2, which is based on well-known models of mosquito-borne diseases (dengue, Zika, etc.). With this vector-host model, we are able to explore the impact of different control strategies (such as vaccination, reducing mosquito abundance and human exposure to mosquito), which could not examined with approaches that fail to incorporate vector-host dynamics.



Fig 2. Flowchart of the yellow fever model. Black arrows represent infection status transition paths, red dashed arrows represent transmission paths and the blue arrow represents the vaccination pathway. Square compartments represent host classes and circular compartments represent vector classes. Red compartments represent infectious classes, and gray compartments are the simulated weekly reported cases (Z_h) and deaths (Y_h).

https://doi.org/10.1371/journal.pntd.0006158.g002

The model applied the following notations. For human host populations, S_h represents the number of susceptible individuals, E_h is the number of individuals exposed to YF but not yet infectious, A_h represents the asymptomatic (i.e., with clinically inapparent symptoms) cases, I_h the severe infectious individuals, and T_h the individuals in the toxic stage. Finally, R_h individuals have either recovered from the disease and/or have been vaccinated (or immunized by vaccination).

For the human host population, there are two main pathways, as seen in Fig 2.

 $H_1) S_h \to E_h \to I_h \to T_h \to R_h$

The susceptible hosts become exposed to YFV by the bites of infectious mosquitoes, harbor the virus (move to the Exposed class) and eventually become infected (move to Infected class), enter the toxic stage (move to Toxic class) and then either eventually recover (move to Recovered class) and remain immune, or in the case of 5-50% in this stage (as specified by the CFR), die from the disease (D_h).

H₂) $S_h \rightarrow E_h \rightarrow A_h \rightarrow R_h$

In this second pathway, susceptible hosts become exposed to YF by the bites of infectious mosquitoes, harbor the virus and eventually become infected but only asymptomatic. The latter usually recover and gain future immunity.

In the above scheme, the asymptomatic (A_h) and severe infectious (I_h) hosts may both infect mosquitoes if they have been bitten, as shown in Fig 2. However, asymptomatic cases have a reduced transmissibility, ψ , when compared to that of a typical severe case. Also, note that individuals in the toxic phase no longer have viremia [1], and therefore cannot be infectious [1, 4, 30].

The vector population has only a single pathway:

 $V_1) S_{\nu} \to E_{\nu} \to I_{\nu}$

As in usual notation, S_{ν} is the number of susceptible mosquitoes. It is assumed that vertical transmission of YFV in the mosquito population is relatively small, and could reasonably be neglected for the purpose of this model [1, 31, 32]. It is also understood that mosquitoes are relatively unaffected by the mosquito-borne viruses [33].

Based on the above descriptions, we formulate an ordinary differential equations (ODEs) system that matches the scheme as illustrated in Fig 2:

$$\begin{cases} S'_{h} = -ab \frac{I_{v}}{N_{h}} S_{h} - v(t - t_{0}) \\ E'_{h} = ab \frac{I_{v}}{N_{h}} S_{h} - \sigma_{h} E_{h} \\ A'_{h} = (1 - \delta)\sigma_{h} E_{h} - \gamma_{h} A_{h} \\ I'_{h} = \delta \sigma_{h} E_{h} - \gamma_{h} I_{h} \\ T'_{h} = \gamma_{h} I_{h} - \kappa_{h} T_{h} \\ R'_{h} = v(t - t_{0}) + \gamma_{h} A_{h} + (1 - \theta) \kappa_{h} T_{h} \\ D'_{h} = \theta \kappa_{h} T_{h} \\ S'_{v} = B_{v}(t) - ac \frac{\psi A_{h} + I_{h}}{N_{h}} S_{v} - \mu_{v} S_{v} \\ E'_{v} = ac \frac{\psi A_{h} + I_{h}}{N_{h}} S_{v} - \sigma_{v} E_{v} - \mu_{v} E_{v} \\ I'_{v} = \sigma_{v} E_{v} - \mu_{v} I_{v} \end{cases}$$
(1)
$$Y_{h}^{(i)} = \int_{\text{week } i} \theta \kappa_{h} T_{h} dt \\ Z_{h}^{(i)} = \int_{\text{week } i} [\theta + \rho \cdot (1 - \theta)] \cdot \gamma_{h} I_{h} dt$$

Here, v(t) represents the vaccination rate at time t, and t_0 is the mean time period from receiving vaccination to acquiring full immunity. $Y_h^{(i)}$ represents the weekly recorded deaths due to YF of the *i*-th week. It is calculated as an integral which effectively sums the weekly number of toxic phase individuals (T_h) who leave or are removed from the toxic class $(\kappa_h T_h)$ over one week. Only a fraction (θ) of the latter dies, where θ is the CFR for severe cases. Similarly the variable $Z_h^{(i)}$ denotes the weekly recorded (observed) cases. This is determined through the term $[\theta + \rho \cdot (1 - \theta)]$ which collects the deaths (through θ) and the non-fatal severe cases via the severe case reporting ratio ρ (more detailed discussion regarding YF case counting can be found in <u>S2 Text</u>). It is assumed that all deaths are reported, and that the fitting procedure can estimate the reporting ratio of severe cases ρ .

In terms of the total host and vector population sizes (N_h and N_ν), the following relations must hold:

$$N_h = S_h + E_h + A_h + I_h + T_h + R_h + D_h = \text{constant}$$
(2)

$$N_v = S_v + E_v + I_v \tag{3}$$

In our model, $N_{\nu} = N_{\nu}(t)$ is time-dependent in a manner that is controlled by the mosquito birth rate $B_{\nu}(t)$ and death rate $\mu_{\nu}(t)$, namely $N'_{\nu} = B_{\nu}(t) - \mu_{\nu}(t) \cdot N_{\nu}$. Following [34, 35], we suppose that

$$N_{\nu}(t) = m(t) \cdot N_h \tag{4}$$

Here the constant $N_h = 6,543,000$ is the number of humans in Luanda province. The parameter m(t) is the time-dependent ratio of the mosquito-to-human populations that needs to be estimated. It is assumed that m(t) is an exponential cubic spline function of time with number of nodes n_m (see Supplementary Information section S1 Text). Nodes are distributed uniformly over the time-domain with values (m_i) that are estimated but restricted to lie between 0 and 20. The range was chosen to reflect reality (where m = 20 implies $\mathcal{R}_0 = 10$, which is beyond the upper bound of \mathcal{R}_0 for YF).

The procedure for modelling human reactive behaviour also relies on the time-dependent parameter m(t). In practice, vector control efforts (i.e., "mosquito-fogging") were case-driven, and fogging was implemented in localities where cases and mortality were reported. Hence it is somewhat reasonable to link mortality to the mosquito-abundance and the transmission rate (see a discussion of this in <u>S8 Text</u>). For a situation in which m(t) is controlled by death rates only, the cubic spline function for m(t) (Eq.4) can be replaced with a simple function based on the YF mortality (<u>S5 Text</u>):

$$m(t) = m_{\text{hase}} + k \cdot \exp\left[-D_{h}(t - t_{\text{hag}})\right].$$
(5)

Here m_{base} is a constant term, k is a parameter controlling the strength of the death-induced human reaction, $D_h(t)$ is the yellow fever deaths of week t and t_{lag} is the lag time for the population response in reaction to mortality levels.

Model parameters. The model is parameterized from prior knowledge of YF, and uses parameter values that are accepted in the literature. Table 1 summarises all model parameters and their ranges. Table 2 summarises parameter values for the different scenarios and model estimates (discussed below).

With regard to parameters for the host population in Eq.1, σ_h^{-1} and γ_h^{-1} represent the host latent and infectious period respectively, with both being approximately 4 days. The latent period also indirectly allows a four-day reporting delay. Symptoms appear when patients leave

Table 1. Summary of parameters.

Parameter	Notation	Value/Range	Unit/Remark	Source
mosquito biting rate	а	0.3-1.0	per vector∙day	[36]
transmission probability from vector to host	Ь	0.10-0.75	per bite	[36]
transmission probability from host to vector	С	0.30-0.75	per bite	[37]
vaccination rate	ν	0-0.043	per day	[2, 8]
host latent period	σ_h^{-1}	3-6	days	[2, 3, 38]
vector latent period	$\sigma_{_{V}}^{-1}$	8-12	days	[4]
non-severe case relative infectivity	ψ	0.1-0.5	Nil	-
host infectious period	γ_h^{-1}	3-4	days	[2, 3]
severe case proportion	δ	15%	Nil	[3, 27]
severe case CFR	θ	0%-50%	Nil	[2-4, 39]
toxic phase duration	κ_h^{-1}	7-10	days	[2, 3]
vector lifespan	μ_{ν}^{-1}	4-35	days	[36, 37]
severe case reporting ratio	ρ	0.01-0.99	Nil	-
initial susceptible host	$S_h.0/N_h$	0.62	fixed, Nil	[8]

https://doi.org/10.1371/journal.pntd.0006158.t001

Table 2. Parameter summary for two scenarios. X.0 denotes X(t = 0), which is the number individuals in X class at the beginning of the study period.

Parameters	Notation	weak	strong	Туре
mosquito biting rate	a (per day)	0.5	0.5	fixed
transition probability from vector to host	<i>b</i> (per bite)	0.4	0.4	fixed
transition probability from host to vector	<i>c</i> (per bite)	0.5	0.5	fixed
host latent period	σ_h^{-1} (days)	4	4	fixed
host infectious period	γ_h^{-1} (days)	4	4	fixed
toxic case duration	κ_h^{-1} (days)	8	8	fixed
vector latent period	$\sigma_{_{_{\mathcal{V}}}}^{_{-1}}$ (days)	10	10	fixed
vector lifespan	μ_{ν}^{-1} (days)	20	20	fixed
severe case proportion	δ	0.15	0.15	fixed
non-severe case relative infectivity	ψ	0.1	0.5	fixed
severe case CFR	θ	0.06	0.06	fixed
number of nodes	n _m	7	7	estimated
severe case reporting ratio	ρ	0.71	0.72	estimated
mean $m(t)$	$\langle m(t) \rangle$	6.34	2.79	estimated
over-dispersion	τ	0.002	0.0045	estimated
initial susceptible host	$S_h.0/N_h$	0.62	0.62	fixed
initial exposed host	$E_h.0/N_h$	3e-07	3.2e-07	estimated
initial non-severe host	$A_h.0/N_h$	3e-07	3.2e-07	estimated
initial severe host	$I_h.0/N_h$	3e-07	3.2e-07	estimated
initial toxic host	$T_h.0/N_h$	3e-07	3.2e-07	estimated
initial recovered host	$R_h.0/N_h$	0.38	0.38	fixed
initial susceptible mosquito	$S_{v}.0/N_{h}$	11.66	3.62	estimated
initial exposed mosquito	$E_{\nu}.0/N_h$	1.66e-06	1.93e-06	estimated
initial infectious mosquito	$I_{v}.0/N_{h}$	1.66e-06	1.93e-06	estimated
mean basic reproductive number	$\langle \mathcal{R}_0 angle$	3.41	2.57	estimated
infection attack rate (%)	AR	0.09	0.15	estimated
maximum log likelihood	MLL	-166.48	-165.4	estimated
Bayesian Information Criterion	BIC	388.9	386.75	estimated

https://doi.org/10.1371/journal.pntd.0006158.t002

the latent class, but are reported only when they leave the infectious class which is a four-day period. The toxic phase duration κ_h^{-1} is set to eight days.

The parameters on mosquitoes were taken from the dengue literature, where mosquito dynamics is also modelled. Since dengue and YF viruses belong to the same family of viruses (i.e., *flaviviridae*) and share the same type of vectors (i.e., *Aedes aegypti*), we follow the practice of previous studies and assume they have similar parameter values. Specifically, parameter values for the mosquito biting rate (*a*), the transmission probabilities (*b*, *c*), and the mosquito lifespan (μ_{ν}^{-1}) were taken from Massad *et al.* [40], as indicated in Table 1. The vector latent period σ_{ν}^{-1} and lifespan μ_{ν}^{-1} were taken as 10 and 20 days respectively.

Vaccination. The term v(t) appearing in the equation for susceptible host dynamics (Eq 1), represents the time-dependent vaccination rate of the host population (blue arrow in Fig 2). It is determined by considering the equation for susceptible dynamics (from Eq 1): $dS_h/dt = -abI_v/N_h - v(t - t_0)$, thus, the rate of vaccinating people is v(t) and the total number of people who would normally be vaccinated by time t is $V(t) = \int_0^t v(x)dx + V_0$. Note that t_0 is the mean time taken for an individual to gain full immunity after being vaccinated. Averaging the data reported by WHO and CDC [2, 3] gives $t_0 = 20$ days. The overall cumulative vaccination for Luanda province, as reported by WHO, is plotted in Fig 1 as a percentage of the total population N_h . That is, the y-axis plots $V(t)/N_h \times 100$. Using this graph and the relation dV(t)/dt = v(t) allows us to reconstruct v(t) which is used in the numerical integration of Eq 1.

The constant V_0 denotes pre-existing immunity of the population at the beginning of the 2015-16 YF outbreak. As the attack rate of YF is typically very small and no major outbreaks occurred in Angola since 1988, we suppose that previously built-up immunity is relatively small and acquired immunity waned significantly over the next 27 years. Nevertheless, it was assumed that at the beginning of each simulation, $V_0 = 38\%$ of the population is already vaccinated to be consistent with WHO estimates. [8]. The 38% includes both the outcome of EPI (WHO's Expanded Program on Immunization) vaccination and immunity remaining from the mass campaign in 1988 (this is incorporated by setting $S_h.0/N_h = 0.62$ as in Table 1). Wu *et al.* [12] assumed that initial immunity was equivalent to 28% vaccination coverage, which under-estimates the WHO data. Results and Discussion section provides more information about YF vaccination doses.

The approach for modelling vaccination is adapted from our previous work on influenza [41], and avoids the need for inclusion of a separate vaccinated compartment in the model which would result in unnecessary additional complexity.

Different YF vaccination intervention scenarios are compared in order to evaluate the effectiveness of the actual national vaccination campaign. The best-fitting model to the data and the actual vaccination coverage will be taken as the "baseline scenario" as experienced in Luanda, which was initiated on February 2, 2016. This will be compared to three other additional hypothetical intervention scenarios.

- Actual vaccination campaign as experienced in Luanda (baseline scenario);
- 60, 120 and \geq 180 days delay of vaccination campaign (hypothetical intervention scenarios).

The total observed cases as well as the total deaths are evaluated by the model for each vaccination scenario.

The 180-day delay period in fact represents a "no-vaccination" scenario. When taking into account the extra 20 days required for vaccination to be effective, anyone vaccinated 180 days after the February 2, 2016, will not gain any protective effect from the vaccination given the observation study period is only 200 days. Thus, any scenario with a delay greater than 180 days is equivalent to a no-vaccination scenario.

Fatality per infection and case-fatality-ratio. Monath *et al.* [4] estimated the fatality per infection of YF for the whole population to be in the range of 3-7.5% [4], and are variable across time and location. In an earlier study, Monath *et al.* estimated the fatality per infection to be in the range of 1-15% for Nigerian villages [42]. For severe YF cases, the Case-Fatality-Ratio (CFR) resulting in death (θ) is 20-50% or higher [4, 43–45], although the CFR is well known to be highly variable, and dependent on particular circumstances.

Given the large proportion of infected but asymptomatic YF cases, the accurate fatality per infection in Luanda cannot be determined without a comprehensive serological study, which to our knowledge has not been undertaken. But the Cases-Fatality Ratio (CFR, cases refer to severe cases) can be immediately approximated as the ratio of confirmed deaths to the confirmed (and probable) cases. For the data of Luanda province, the CFR is approximately 7.76%, and thus substantially lower than 20-50%. (Similar low estimates were noted by the WHO reports throughout the epidemic.) Moreover, because of reporting errors, we would expect the CFR to be even lower than this empirical estimate.

In our study, we choose CFR = 6% for our main simulations. But we have also carefully explored other possibilities. For example, in S2 Text, we run our fitting procedure to actually estimate the CFR given the data and find CFR = 4%. We also discuss what might be expected if the mortality data is under-reported (see S7 Text). More data and research is needed to gain a better understanding of the CFR and to check whether and how it changes over the study period.

Finally, we note that in our model, the fatality per infection is given by $\delta \cdot \theta$, where the CFR = θ , while δ is the proportion of severe cases. In this work, we consider either fixing the CFR to a value considered realistic, or inferring the CFR from the data itself, as an extra parameter.

Asymptomatic infections. As reported by the CDC, "asymptomatic or clinically unapparent infection is believed to occur in most YFV infections" [3]. A case is defined as an asymptomatic infection only if it is confirmed strictly to have no symptoms, but is nevertheless found infectious as confirmed by RNA or serological tests. It was previously believed that 6 out of 7 YF infections could be asymptomatic [26]. Recently, a meta-analysis based on 11 studies suggested that the asymptomatic ratio should be 55% and mild cases 33% (without jaundice), the rest 12% are severe cases [27]. As mentioned, to simplify presentation all mild cases (without jaundice) were considered to belong to the asymptomatic class. Two scenarios were examined:

1. 85% asymptomatic ($\delta = 15\%$) and weak infectivity ($\psi = 0.1$)

2. 85% asymptomatic ($\delta = 15\%$) and strong infectivity ($\psi = 0.5$)

The proportion of 85% for asymptomatic infections based on [27]. Scenarios 1 & 2 differ only in terms of their weak ($\psi = 0.1$) or strong ($\psi = 0.5$) infectivity. The results for scenarios 1 & 2 (with fixed CFR = 6%) are presented in the main text, while results of flexible CFR can be found in S2 Text.

Basic reproduction number. Following the next generation matrix method [46, 47], the basic reproduction number for the above model equations (see Eq.1) was calculated as:

$$\mathcal{R}_{0} = \sqrt{\left[\psi \cdot (1-\delta) + \delta\right] \cdot \frac{a^{2}bcm}{\gamma_{h}} \cdot \frac{\sigma_{v}}{\mu_{v}(\sigma_{v} + \mu_{v})}} \tag{6}$$

as derived in S3 Text. $\mathcal{R}_0(t)$ is calculated as a function of time, by assuming the mosquito-tohuman population ratio to be m = m(t) (see Eq 4), which varies with time in a manner that may be estimated by the model fitting procedures.



Fig 3. Model fitting results under two scenarios: Scenario 1 (weak infectivity with $\psi = 0.1$) in panels (a,b) and scenario 2 (strong infectivity with $\psi = 0.5$) in panels (c,d). Black line with circles denote reported cases (in (a), in the form of square-root) and reported deaths (in (b), in the form of square-root), and red line denotes model simulation median. Blue dashed line denotes the fitted basic reproduction number, $\mathcal{R}_0(t)$, and the green dashed line shows the calculated host susceptible proportion, S(t) (or $S_h(t)/N_h$ in the model Eq 1). Shaded region represents 95% bound of 1,000 model simulations. Vertical dashed line indicates the start date of the vaccination campaign. Inset panel shows BIC as a function of the number of nodes (n_m) . The lowest BIC is attained at $n_m = 7$ in both scenarios, which is used in the main panel. Parameter values are listed in Table 2.

https://doi.org/10.1371/journal.pntd.0006158.g003

When the population is not fully susceptible, it is a common practice to make use of \mathcal{R}_{eff} , the "effective \mathcal{R}_0 " given by $\mathcal{R}_{\text{eff}}(t) = \mathcal{R}_0(t) \cdot S_h(t)/N_h$, to describe the ability of a virus to invade the host population [48]. \mathcal{R}_{eff} incorporates both the changes in the intrinsic ability of the virus, the characteristics of the mosquito vector as well as the availability of susceptible human hosts. We note that the vaccination campaign can only reduce the availability of human susceptibles but not the intrinsic transmissibility of the virus. Vector control (e.g., mosquito fogging) can reduce the transmissibility of the virus, by reducing its vector of transmission. In Fig 3, we plot both $\mathcal{R}_0(t)$ and S(t). From these, it is simple to obtain $\mathcal{R}_{\text{eff}}(t)$. In Fig 4, we compare $\mathcal{R}_0(t)$ with a reconstructed transmission rate (see next section).

Plug-and-play inference framework

The YF outbreak in Luanda is modeled as a Partially Observed Markov Process (POMP) and makes use of the Iterated Filtering and plug-and-play likelihood-based inference frameworks to fit the data [34, 49, 50]. These are modern state-of-the-art statistical methodologies developed for fitting complex epidemiological datasets. The Maximum Likelihood Estimate (MLE) for model parameters is calculated using R package "POMP" [51].

Bayesian Information Criterion (BIC) is employed as a criterion for model comparison, and quantifies the tradeoff between the goodness-of-fit of a model and its complexity [52]. The simulations made use of the Euler-multinomial integration method with the time-step fixed to be one day [49, 53].



https://doi.org/10.1371/journal.pntd.0006158.g004

The model is first fitted to the observed YF cases and deaths, given knowledge of the true vaccination coverage. The mosquito abundance is assumed to be unknown but time-dependent, and is reconstructed. We allow the basic reproduction number of our model to be time-dependent, given that the mosquito abundance is not fixed and human behavior can impact $\mathcal{R}_0(t)$ and change over the study period.

The parameter fitting and inference process are carefully checked, thereby giving high confidence that the fits of the observed time-series are accurate for reasons that are consistent with the true underlying epidemiological processes rather than artificial model over-fitting. We conducted tests to find the best-fit model. For each asymptomatic scenario, we studied 10 different values of n_m (degrees of freedom in the m(t)), and compared them with BIC. BIC quantifies the trade off of the goodness of fitting of the model and the complexity of the model penalizing models with more variables. A smaller BIC implies a better-fit model. For the bestfit model, the profile of maximum log likelihood was calculated as a function of the reporting ratio (see <u>S2 Text</u> in SI for further details). The profile found is always a reasonably smooth function. The model was run 1,000 times with the estimated parameters, and the median of the model simulation matched the reported weekly cases. Therefore, we can be confident that the maximization of model's log likelihood converged and the estimation is consistent. The simulated weekly reported cases Z_t are modelled by Eq 1. The corresponding weekly observed cases, C_t , as given by the WHO, are assumed to follow a Negative-Binomial (NB) distribution as

$$C_t \sim \text{NB}\left(n = \frac{1}{\tau}, p = \frac{1}{1 + \tau Z_t}\right)$$
 with mean : $\mu_t = Z_t$, (7)

where τ denotes an over-dispersion parameter that needs to be estimated.

The weekly observed deaths, D_t , and the corresponding weekly simulated deaths, Y_t , are similarly related. Finally, the overall log-likelihood function, l, is given by

$$l(\Theta|C_1, \dots, C_N; D_1, \dots, D_N) = \sum_{t=1}^T \ln [L_t^{(C)} \cdot L_t^{(D)}],$$
(8)

where Θ denotes the parameter vector under estimation, and $L_t^{(C)}$ and $L_t^{(D)}$ are the probability measurement functions associated with C_t vs. Z_t , and D_t vs. Y_t , respectively. T denotes the total number of weeks during the study period.

The confidence intervals (C.I.) of parameters are estimated based on parameters ranges in Table 1, using the method of profile likelihood confidence intervals [35, 49]. This is demonstrated in S2 Text for the severe case reporting rate ρ . Parameter estimation and statistical analysis are conducted using R (version 3.3.3).

Sensitivity analysis

The Partial Rank Correlation Coefficients (PRCCs) are adopted for the model's sensitivity analysis [34]. Firstly, 1,000 random samples are taken for each model parameter from uniform distributions with parameter ranges as set out in Table 1. After that, for every random parameter sample set, the YF model was simulated to obtain the target biological quantities (e.g., \mathcal{R}_0 and total number of deaths in this study). Finally, PRCCs were calculated between each parameter and target biological quantities.

Results and discussion

Model fitting

The results for the best-fitting model under the two scenarios (i.e., weak and strong infectivity scenarios) are shown in Fig 3. The model simulation median (of 1,000 simulations) of YF cases in Luanda is plotted in red and matches well the observed patterns seen in weekly reported cases, both before and after the national vaccination campaign. The two scenarios (for asymptomatic infectivities) both model the data with almost the same goodness-of-fit with a Δ BIC \approx 2 (see S2 Text for the simulation results of strong infectivity scenario, i.e., scenario 2). That is, the observed and model time series are not significantly different for the two levels of infectivity [52]. As such the infectivity of asymptomatic cases cannot be accurately inferred from these data sets.

In <u>Table 2</u>, the over-dispersion τ , is notably small, which indicates the measurement model is close to a Poisson distribution (i.e., minor over-dispersion in measurement noise). This implies the reporting efforts (i.e., reporting ratios) were reasonably stable over time.

The analysis estimated a mean $\mathcal{R}_0 \approx 2.6$ -3.4 and an attack rate of the whole period to be 0.09-0.15% (% population infected by YF) from December 2015 to August 2016. Our estimated initial and upper bound \mathcal{R}_0 are in line with previous studies.

Asymptomatic cases were not reported, and they might be considered as a completely hidden variable. However, if the number of asymptomatic cases is very large (e.g., if the



Fig 5. Simulation results of scenario 1 under three deferred vaccination campaign scenarios: 60 day delay in panels (a,b), 120 day delay in panels (c, d) and 180 day delay in panels (e,f). Black line with circles denote reported cases (e.g., in (a), in the form of square-root) and reported deaths (in (b), in the form of square-root), red line denotes model simulation median and blue dashed line is the fitted basic reproduction number, \mathcal{R}_0 . Shaded region represents 95% range of 1,000 simulations. The vertical dashed line represents initiation of the vaccination campaign. The number of nodes, $n_m = 7$, is adopted.

https://doi.org/10.1371/journal.pntd.0006158.g005

asymptomatic-to-symptomatic ratio is 6:1 or 7:1) with a weak infectivity but full immunity, this will indirectly slow down the transmission of YF in the later stages, due to herd immunity built up by these silent asymptomatic cases. If their infectivity is strong, this will increase the difficulty to control the outbreak.

The model simulations of weekly deaths also fit the observed data well over the period of the main epidemic until the end of April 2016. While the simulated median (red line) does not predict the two relatively small and erratic peaks at the beginning of June and end of July, nevertheless they fit reasonably within the 95% bounds. (Note that similar peaks in death numbers appear in the delayed vaccination scenarios Fig 5b, 5d and 5f, where case numbers are higher. See next section.)

The observed YF deaths are relatively "noisy" compared to the continuously observed YF cases (see red bars versus green dotted line in Fig 1), which might be due to the lower case numbers involved or possibly spatial variation of the YF CFR. The same holds for individual model simulations. The total number of deaths was only 6% out of all reported cases (i.e., CFR = 6%), and 71% of the deaths appeared during the first wave. Although we cannot fit the final erratic mortality waves with high accuracy, our estimate of the total number of deaths is still a very good approximation. As can be seen in Table 3, the model's simulated cumulative death toll matches well with the observed death toll.

Parameter estimates including the basic reproduction number \mathcal{R}_0 , mean mosquito-tohuman ratio $\langle m(t) \rangle$, and disease attack rate are listed in <u>Table 2</u>. Our estimated mean $\mathcal{R}_0 \approx 3.0$ with excursions to $\mathcal{R}_0 \approx 6.0$ matches well other studies in the literature (see <u>Introduction</u>). The very low attack rate is an outcome of the prompt and efficient control measures by the Angolan government [2, 6, 8].

Scenario	Total reported cases	Total deaths	
Observed	941	73	
Baseline model	1026 [540, 1797]	77 [35, 139]	
60 days delay	3143 [1604, 5584]	233 [119, 411]	
120 days delay	5450 [2751, 9611]	400 [203, 724]	
180 days delay	6242 [3139, 10919]	444 [226, 787]	

Table 3. Impacts of vaccinat	on campaign delay under	scenario 1: Weak infectivity
------------------------------	-------------------------	------------------------------

https://doi.org/10.1371/journal.pntd.0006158.t003

The estimated mean mosquito-to-human ratio, $\langle m(t) \rangle$, is in line with previous work by Gao *et al.* (see [54]). The estimated reporting ratio for severe cases, $\rho = 70\%$, is reasonable, given the easily recognizable symptoms (jaundice), and the control effort by the government, which managed to push the vaccination coverage to more than 90% of its population within a very short period of time.

Association between the spread of YF and local climatic factors has been discussed frequently in previous studies [55, 56]. As such, we explored the possibility that local temperature and rainfall are potential factors that consistently influence the long-term transmission dynamics. Temperature was found to have no significant effect while the effects of rainfall were significant but quantitatively mild in a first analysis. The possible reasons could be: i) local precipitation is relatively minor during the study period but concentrated in March, and the weather is continuously hot and dry; and ii) duration of the outbreak is short and other factors (control measures and human reaction) played a more prominent role. Nevertheless, in what follows, we consider the possibility that the March El Niño rainfall patterns played an important role which is difficult to untangle from our analysis. A discussion is given in <u>S6</u> <u>Text</u>.

Oscillations in $\mathcal{R}_0(t)$ **and human reactive behaviour.** For the two scenarios, the analysis revealed that $\mathcal{R}_0(t)$ oscillated in the interval roughly [1.0,8.5] over the 37 weeks with mean $\mathcal{R}_0 \approx 3.4$ of weak infectivity, and $\mathcal{R}_0 \approx 2.6$ for strong infectivity. The estimated \mathcal{R}_0 (see blue dashed line in Fig 3) is approximately from 5.0-8.5 (see CI estimate of $\mathcal{R}_0(t)$ on S4 Text), which is in line with Wu *et al.*'s [12, 13] analysis of the early stage of the epidemic (\mathcal{R}_0 is from 5.2-7.1).

We now show that the oscillation in $\mathcal{R}_0(t)$ is a robust feature of the dataset based on Fine and Clarkson's [57] effective and well known methodology. Fine and Clarkson plotted $Q(t) = C_{t+1}/C_t$ (i.e., next week's reported cases divided by this week's—as used in many classic studies of childhood infectious diseases), as a function of time *t*. This index is usually regarded as the transmissibility of human-to-human infectious diseases but can be similarly exploited for vector-borne diseases. In Fig 4, we find strong waves in Q(t) (the solid green line) that match the oscillations observed in $\mathcal{R}_0(t)$ (the dashed blue line). This suggests that the waves observed in \mathcal{R}_0 are most likely a feature embedded in the data, and thus not artificial.

A closer examination of Fig 3 helps explain the oscillations in $\mathcal{R}_0(t)$ and their implications. A few weeks after the initiation of the vaccination program in February 2016, the YF epidemic is seen to curtail its rapid growth and then diminish over March. Unusually, the epidemic does not rapidly crash to extinction as the usual (SIR) epidemic curve would predict. Rather, YF cases reduced gently over the next five months in a plateau from April to September. This behavior is uncharacteristic of the standard SIR epidemic model, and appears to be an outcome of the oscillations observed in the time-series estimates of $\mathcal{R}_0(t)$ (see blue dashed line in Fig 3). Thus the epidemic begins to rapidly decline and turns-around only after \mathcal{R}_0 has reduced considerably (i.e., after the peak in $\mathcal{R}_0(t)$ in January). A similar decreasing trend in \mathcal{R}_0 over Feb-Mar 2016 was also noted by [12]. This implies that although vaccination was important for the main epidemic's demise, the February decline in \mathcal{R}_0 was also likely to have played an important role. Interestingly, the sustained period of cases in April (rather than the expected drop to disease extinction) occurs when \mathcal{R}_0 increases again. Since theoretically, \mathcal{R}_0 should remain unaffected by vaccination, we suppose the changes in \mathcal{R}_0 are likely due to changes in other factors (e.g., control measures and behavior, or possibly weather patterns) that may influence the vector population and its transmission.

Human reaction in responding to YF deaths (i.e., the "recent-death-driven" human reaction) could also be involved in this complex wave-like behavior of $\mathcal{R}_0(t)$ (see <u>S8 Text</u>), which could be due to the interference of multiple disease control strategies. The control efforts on mosquito-eradication increased because (or follows) the number of YF cases and deaths increased. After the national vaccination campaign began in the first week of February, 2016, and reported death decreased, the mosquito control effort may have also slowed down (possibly due to limited funding and resources). This is biologically reasonable, and could be due to protective behavioral changes such as usage of insecticide, mosquito repellent, movement restrictions, cordon sanitaire and general vector control. Thus, we hypothesize that the human behaviour (e.g., usage of mosquito insecticides, repellents and nets) in response to cases and deaths were witnessed during Luanda's YF epidemic, of which reduced disease transmission followed high mortality and vice versa. Such a process would drive the YF case numbers to follow dynamics that differ substantially from the standard SIR epidemic curve, and possibly even induce waves. Similar phenomena were reported previously for the 1918 influenza pandemic [58, 59]. Therefore, we hypothesize that human behavior could correspond to the delay from YF deaths to the transmission rate for the next YF wave.

As mentioned in the Methods section, when modelling reactive behaviour we replace the cubic spline function (with 7 nodes) for m(t) (Eq 4) with a simple function based on the YF mortality. When we do this, almost identical results are obtained (see S5 Text). This simply means that the death data are an excellent proxy for \mathcal{R}_0 (since m(t) directly drives \mathcal{R}_0 (see Eq 6).

Generation time. The generation time (GT, i.e., the time between two successive infections) is supposed to be responsible for the delay from YF transmission waves, then, to YF cases and, finally, to YF deaths. The generation time equals the sum of latent and infectious periods of the host [60], and the sum of latent and infectious periods of the mosquito. However, mosquitoes have a short lifespan and could die before the loss-of-infectiousness of YF. Precisely, the life of adult (female) mosquito includes three parts: pre-infection, latent and infectious period. In this work, we fix the mean lifespan (L_{ν}) of mosquito to be 20 days, and the mean latent period to be 10 days. For the pre-infection period, we use the formula of mean age at infection (for typical childhood infections): $L_{\nu}/(1 + \mathcal{R}_0)$ [48]. If we fix $\mathcal{R}_0 = 3.0$ (see Table 2), the mean age at infection of mosquito is 5 days. Thus, the sum of latent and infectious periods of mosquito will be 20 - 5 = 15 days. The sum of latent and infectious periods of human is 4 + 4 = 8 days. Thus, the GT of YF is 23 days, which explains the time delay between the maxima of $\mathcal{R}_0(t)$ and the maxima of reported weekly cases. GT will be between 18 and 28 days, which is in line with [13].

What caused the second wave in the oscillatory pattern of $\mathcal{R}_{0}(t)$?

For most years in Luanda, the rainy season is between November-May but the most accumulation of rain occurs in March-April [61]. The year 2016 was an El Niño year and it brought dramatic and unpredictable flooding events especially in the March-April period, thereby leading to conditions ideal for growth in mosquito populations. As in 1971, YF outbreak in Luanda, local water-storage containers (mainly the larger ones) serving the community but also in most homes, accounted for 85% of the *Ae. aegypti* larval breeding sites [61, 62]. As vividly described by Moreira [63]: "*The 2016 outbreak coincided with unusually heavy rains and a severe El Niño weather pattern. We are also suffering from an economic crisis and poor sanitary conditions. All these factors created a fertile environment for an increase in the mosquito population. The outbreak reached its peak in February and has been declining since (i.e., population numbers, not \mathcal{R}_0). We have much more vaccine now (in September 2016) than we had earlier in the epidemic. The response interventions are involving communities successfully. The dry season arrived in May (2016) and since then the mosquito population has diminished."*

Thus after the peak of the YF outbreak had passed, and the vaccination program was in progress, the local March-April El Niño rains were enhancing mosquito breeding conditions. It is surprising that simultaneously one of Luanda's largest malaria epidemics ever was underway ("*During the first quarter of 2016, the number of cases of malaria increased dramatically to 1,531,629, up from 980,192*" [64]). We suggest that these conditions may also be responsible for the unusual but robust second wave that is observed in the time series of $\mathcal{R}_0(t)$.

Since we only model a single province, Luanda, and the YF transmission spread relatively rapidly throughout the province, we might assume the effects of spatial heterogeneity are likely to be minimal. However, we do not possess sufficiently detailed data to perform a careful analysis of spatial effects, and effects at the micro-scale may be important as in other diseases such as dengue [65]. Kraemer *et al.* [13] have discussed the importance of spatial effects for YF over all provinces of Angola, and there is a possibility that geographic waves generated from surrounding provinces could play some part in the appearance of multiple YF waves in Luanda province. Hence future work and more comprehensive data are needed to examine these possibilities.

Vaccination

There are many possible ways to evaluate the effects of a delayed vaccination campaign when compared to the baseline scenario that was implemented in practice in Luanda. The approach followed here is to simply delay the exact same baseline scenario (in terms of doses per week) by a fixed time interval until the end of the observation period arrives. It is difficult to extend beyond the observation period without introducing an unacceptable rate of errors. This can be seen in the large confidence intervals for \mathcal{R}_0 towards the end of the observation period (see S4 Text).

The results of 60, 120, and 180 days delay of the vaccination campaign for the 2016 yellow fever outbreak are presented in Fig 5 for the scenario 1 ($\psi = 0.1$). The total reported cases and total deaths are calculated for four vaccination scenarios (including the baseline) and outcomes are listed in Table 3. The 180-day delay is included, because it gives an impression of what might happen when vaccination is unavailable, as mentioned.

The baseline scenario (actual vaccination or 0-delay) results in an estimated 73 deaths associated with YF in the study period, which matches the observed number. With a 60-day delay to the vaccination roll-out, YF deaths saved were 2.2-fold of the observed number (see Fig 5a and 5b). With a 120-day delay, the YF death saved were 4.5-fold of the observed number (see Fig 5c and 5d). With a 180-day delay, YF deaths saved were 5.1-fold of the observed number (see Fig 5c and 5d). With a 180-day delay, YF deaths saved were 5.1-fold of the observed number (see Fig 5e and 5f). The latter result is a good approximation to what might have occurred if there were no vaccination campaign in Luanda up to August 2016. All of these results show that delaying the vaccination campaign would have greatly enhanced the epidemic in terms of infectious cases and mortality. We also investigated the "vaccination delay" situation under different scenarios (see S2 Text), and found our main results of "deaths prevented" largely

Delay	Vaccinated population since outbreak	Doses usage (person dose)
0 day	93% - 38% = 55%	3,598,650
60 days	91% - 38% = 53%	3,467,790
120 days	90.5% - 38% = 52.5%	3,435,075
180 days	54% - 38% = 16%	1,046,880

Table 4. Vaccine doses used during yellow fever outbreak (from Dec 5, 2015 to Aug 15, 2016) under four vaccination delay situations.

https://doi.org/10.1371/journal.pntd.0006158.t004

holds. In addition, we also considered the scenario of "what if deaths were under reported" (i.e., there was a constant proportion of YF deaths not reported, see <u>S7 Text</u> for details), we report our main results are also robust.

A clear feature of the simulated outcomes with delayed vaccination (red lines in Fig 5), is the noticeable second wave of YF cases and deaths that appear. This feature becomes even more prominent in a situation of no vaccination (see Fig 5e and 5f). Returning to Fig 1, we also see strong signs from the observed time series of YF in Luanda, that the outbreak may have indeed occurred over two waves. Hence, even with Luanda's large-scale vaccination campaign, the multiple-wave feature is noticeable in the observed time series, which implies considerable fluctuations in the driving force (\mathcal{R}_0) or other factors.

Vaccine usage. As discussed in the main text, we examined scenarios when an identical vaccination scheme to that observed in Luanda was delayed by 60 and 120 days. Here we show the number of doses given in these delayed schemes is almost identical to the baseline scheme varying at most by 5% (for delay of 120 days, 3.44 million doses). The 180-day delay is included because, as explained in the text, it is so late that it is actually a no-vaccination scenario (see below).

The calculations are based on Luanda having a population size of 6,543,000, with 38% of the population having immunity before the epidemic (November 2015). The baseline scenario characterized what happened in practice between December 2015 and August 2016. By August 2016, the vaccine-dose usages are summarized for all four scenarios in Table 4. The total reported observed cases as well as total deaths are evaluated by the model for each different vaccination scenario.

A 180-day delay period is considered to be a no-vaccination scenario. When taking into account the extra 20 days required for the vaccine to be effective, anyone vaccinated 180 days after December 2015, will not gain any protective effect from the vaccination given the observation study period is only 200 days. The impact of the vaccination goes beyond the study period. Anyone vaccinated in the last twenty days will change their status from susceptible to recovered after 15 August 2016 (i.e., the end of the study period).

Sensitivity analysis

Results of sensitivity analysis are presented in Fig 6 and indicate how model parameters impact the basic reproduction number \mathcal{R}_0 and the death toll. \mathcal{R}_0 is most sensitive to vector biting rate (*a*) and the vectors' lifespan (μ_{ν}^{-1}), indicating the importance of the mosquitoes' role in disease transmission. The total deaths are considerably sensitive to the proportion of severe cases (δ), the case-fatality rate of severe cases (θ) and the initial number of susceptibles (i.e., the ratio $S_h \cdot 0/N_h$).

Conclusion

Using modern likelihood based statistical inference techniques, it was possible to fit a vectorhost epidemic model successfully to the surveillance data collected for the YF outbreak in



Fig 6. The Partial Rank Correlation Coefficients (PRCCs) of basic reproduction number (panel (a)) and total deaths (panel (b)) with respect to model parameters. S_h .0 denotes the initial susceptible ratio (S_h .0/ N_h). The black dots are the estimated correlations and the bars represent the 95% C.I.s. The ranges of parameters are given in Table 1.

https://doi.org/10.1371/journal.pntd.0006158.g006

Luanda, Angola in 2016. We were thus able to assess the success of the vaccination campaign as rolled out in Luanda. While there were in reality 73 deaths reported over the 37-week study period, the model showed that the vaccination campaign saved from death 5.1-fold of observed deaths and prevented from illness 5.8-fold of observed cases, over the study period, and no doubt many more if we were to extrapolate beyond the study period. This was determined by simulating Luanda's YF outbreak in the absence of any vaccination. The national vaccination campaign was also found to be timely, in that delaying the availability of the vaccination any further would have greatly enhanced the epidemic in terms of number of YF cases and mortality.

The change in the number of YF cases over time in Luanda suggests the possibility that the outbreak occurred in two waves over the 37-week study period. The modelling and sensitivity analysis demonstrated that this is a robust feature (see S5 Text), which would have become far more prominent had the vaccination campaign been reduced in intensity. The appearance of waves implies that \mathcal{R}_0 must oscillate to some degree in time. Reconstruction of the underlying dynamics reveals that \mathcal{R}_0 is strongly out of phase with mortality, so that \mathcal{R}_0 decreases when the number of deaths increase, and vice versa. Thus we hypothesize that the high death rates and number of cases influenced Luanda's population behavioral response which in turn led to some reduction of disease transmission during the years of high mortality. Behavioral responses may typically involve using more insecticide, mosquito repellent, insecticide-treated bednets and broader vector-control programs, as outlined in S8 Text. In Luanda, it also involved cordon sanitaire, and movement restriction with the aim of reducing transmission through the wider population [66].

Such behavioral changes are able to modulate the basic reproduction number, which in turn can lead to waves in the YF case numbers. A similar phenomenon was reported for the deadly pandemic influenza (e.g., 1918 influenza pandemic with a fatality rate (per infection) 2% and an attack rate 1/3) [58, 59] but never in mosquito-borne disease since either the CFR or the AR is

typically low. Moreover, we showed how a simpler model that explicitly incorporated human behavior reproduces the observed data in Fig 1 (see S5 Text). This may be the first example of mortality-driven basic reproduction number in a mosquito-borne disease outbreak. While this possibility appears to hold in other epidemiological contexts (e.g., Spanish flu [58, 59]), it would be beneficial to check this further for vector-host systems. In the case of Luanda's YF outbreak, \mathcal{R}_0 is likely to have also been affected by the sporadic but heavy El Niño rainfall, which in turn could influence mosquito population numbers. Such a process could occur even if there is no visible long-term correlation between climate (rainfall) and the vector dynamics.

The modelling approach described here provides a basis for future vaccination campaign evaluations. Since the YF mortality appeared to lead to oscillations in the basic reproduction number (\mathcal{R}_0), this possibility should be considered in the development of short-term prediction tools of the spread of YF. The general approach should be of benefit in mitigating the spread and impact of YF outbreaks in the future.

Supporting information

S1 Text. Fitting *m*(*t*) with BIC. (PDF)

S2 Text. Different model scenarios. (PDF)

S3 Text. Basic reproduction number. (PDF)

S4 Text. Estimate the confidence interval of $\mathcal{R}_0(t)$. (PDF)

S5 Text. Underlying oscillation in basic reproductive number. (PDF)

S6 Text. Impacts of temperature and rainfall. (PDF)

S7 Text. Assuming deaths were under-reported (CFR = 15%). (PDF)

S8 Text. Impact of vector control and possible impact of climate. (PDF)

Author Contributions

Conceptualization: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Data curation: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Formal analysis: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Funding acquisition: Daozhou Gao, Daihai He. Investigation: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Methodology: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Project administration: Daozhou Gao, Daihai He. Resources: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He. Software: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He.

Supervision: Lewi Stone, Daozhou Gao, Daihai He.

Validation: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He.

Visualization: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He.

Writing - original draft: Shi Zhao, Lewi Stone, Daozhou Gao, Daihai He.

Writing - review & editing: Lewi Stone, Daozhou Gao, Daihai He.

References

- Monath TP, Vasconcelos PF. Yellow fever. Journal of Clinical Virology. 2015; 64:160–73. <u>https://doi.org/10.1016/j.jcv.2014.08.030</u> PMID: 25453327
- World Health Organization. Yellow Fever Fact Sheet. http://www.who.int/mediacentre/factsheets/ fs100/en/; accessed on Mar, 2017.
- Centers for Disease Control and Prevention. Yellow Fever. <u>https://www.cdc.gov/yellowfever/;</u> accessed on Mar, 2017.
- Monath TP. Treatment of yellow fever. Antiviral Res. 2008; 78:116–24. https://doi.org/10.1016/j. antiviral.2007.10.009 PMID: 18061688
- Barrett AD. Yellow Fever in Angola and Beyond–The Problem of Vaccine Supply and Demand. N Engl J Med. 2016; 375(4):301–3. https://doi.org/10.1056/NEJMp1606997 PMID: 27276108
- 6. Woodall JP, Yuill TM. Why is the yellow fever outbreak in Angola a "threat to the entire world" Int J Infect Dis. 2016; 48:96–7. https://doi.org/10.1016/j.ijid.2016.05.001 PMID: 27163382
- 7. International Society for Infectious Diseases. Program for Monitoring Emerging Diseases (ProMED). http://www.promedmail.org/; accessed on Mar, 2017.
- 8. African Health Observatory (AHO). Publications website, Health Topic: Yellow Fever. Yellow fever outbreak in Angola Situation Reports. http://www.afro.who.int/publications; accessed on Jul, 2017.
- Imperial College, London. Yellow fever Immunization Coverage across Africa. https://polici.shinyapps. io/yellow_fever_africa/; accessed on Mar, 2017.
- African Health Observatory (AHO). Yellow fever outbreak in Angola Situation Report 26 Feb 2016. http://www.afro.who.int/sites/default/files/2017-06/sitrep_ang_feb_26_2016.pdf; accessed on Jul, 2017.
- 11. Anderson RM, May RM, Anderson B. Infectious diseases of humans: dynamics and control. Oxford: Oxford university press; 1992.
- Wu JT, Peak CM, Leung GM, Lipsitch M. Fractional dosing of yellow fever vaccine to extend supply: a modelling study. Lancet. 2016; 388:2904–2911. https://doi.org/10.1016/S0140-6736(16)31838-4 PMID: 27837923
- Kraemer MU, Faria NR, Reiner RC Jr, Golding N, Nikolay B, Stasse S, et al. Spread of yellow fever virus outbreak in Angola and the Democratic Republic of the Congo 2015–16: a modelling study. Lancet Infect Dis. 2017; 17:330–338. https://doi.org/10.1016/S1473-3099(16)30513-8 PMID: 28017559
- Germann TC, Kadau K, Longini IM, Macken CA. Mitigation strategies for pandemic influenza in the United States. Proc Natl Acad Sci U S A. 2006; 103(15):5935–40. https://doi.org/10.1073/pnas. 0601266103 PMID: 16585506
- Mills CE, Robins JM, Lipsitch M. Transmissibility of 1918 pandemic influenza. Nature. 2004; 432 (7019):904–6. https://doi.org/10.1038/nature03063 PMID: 15602562
- White LF, Wallinga J, Finelli L, Reed C, Riley S, Lipsitch M, et al. Estimation of the reproductive number and the serial interval in early phase of the 2009 influenza A/H1N1 pandemic in the USA. Influenza Other Respir Viruses. 2009; 3:267–76. https://doi.org/10.1111/j.1750-2659.2009.00106.x PMID: 19903209
- Lessler J, Reich NG, Cummings DA. Outbreak of 2009 pandemic influenza A (H1N1) at a New York City school. N Engl J Med. 2009; 361(27):2628–36. https://doi.org/10.1056/NEJMoa0906089 PMID: 20042754
- Towers S, Patterson-Lomba O, Castillo-Chavez C. Temporal variations in the effective reproduction number of the 2014 West Africa Ebola outbreak. PLoS currents. 2014; 6. https://doi.org/10.1371/ currents.outbreaks.9e4c4294ec8ce1adad283172b16bc908 PMID: 25642357

- Gomes MF, y Piontti AP, Rossi L, Chao D, Longini I, Halloran ME, Vespignani A. Assessing the international spreading risk associated with the 2014 West African Ebola outbreak. PLoS currents. 2014; 6. https://doi.org/10.1371/currents.outbreaks.cd818f63d40e24aef769dda7df9e0da5 PMID: 25642360
- Fisman D, Khoo E, Tuite A. Early epidemic dynamics of the West African 2014 Ebola outbreak: estimates derived with a simple two-parameter model. PLoS currents. 2014; 6.
- Nishiura H, Chowell G. Early transmission dynamics of Ebola virus disease (EVD), West Africa, March to August 2014. Euro Surveill. 2014; 19(36): 20894. https://doi.org/10.2807/1560-7917.ES2014.19.36. 20894 PMID: 25232919
- 22. De Cock KM, Monath TP, Nasidi A, Tukei PM, Enriquez J, Lichfield P, et al. Epidemic yellow fever in eastern Nigeria, 1986. Lancet. 1988; 1:630–3. https://doi.org/10.1016/S0140-6736(88)91425-0 PMID: 2894558
- Vasconcelos PF, Rodrigues SG, Degallier N, Moraes MA, da Rosa JF, da Rosa ES, et al. An epidemic of sylvatic yellow fever in the southeast region of Maranhao State, Brazil, 1993–1994: epidemiologic and entomologic findings. Am J Trop Med Hyg. 1997; 57:132–7. <u>https://doi.org/10.4269/ajtmh.1997.57</u>. 132 PMID: 9288803
- Nasidi A, Monath TP, DeCock K, Tomori O, Cordellier R, Olaleye OD, et al. Urban yellow fever epidemic in western Nigeria, 1987. Trans R Soc Trop Med Hyg. 1989; 83:401–6. https://doi.org/10.1016/0035-9203(89)90518-X PMID: 2617590
- Monath TP, Nasidi A. Should yellow fever vaccine be included in the expanded program of immunization in Africa? A cost-effectiveness analysis for Nigeria. Am J Trop Med Hyg. 1993; 48:274–99. https:// doi.org/10.4269/ajtmh.1993.48.274 PMID: 8447531
- Monath TP, Craven RB, Adjukiewicz A, Germain M, Francy DB, Ferrara L, et al. Yellow fever in the Gambia, 1978–1979: epidemiologic aspects with observations on the occurrence of orungo virus infections. Am J Trop Med Hyg. 29:912–28. https://doi.org/10.4269/ajtmh.1980.29.912 PMID: 7435793
- Johansson MA, Vasconcelos PF, Staples JE. The whole iceberg: estimating the incidence of yellow fever virus infection from the number of severe cases. Trans R Soc Trop Med Hyg. 2014; 108:482–7. https://doi.org/10.1093/trstmh/tru092 PMID: 24980556
- 28. MuleSoft. City Population.https://www.citypopulation.de/; accessed on Mar, 2017.
- World Health Organization. Emergencies preparedness, response. Yellow fever Case definitions for public health surveillance. 18 May 2015. http://www.who.int/csr/disease/yellowfev/case-definition/en/
- Akondy RS, Johnson PL, Nakaya HI, Edupuganti S, Mulligan MJ, Lawson B, et al. Initial viral load determines the magnitude of the human CD8 T cell response to yellow fever vaccination. Proc Natl Acad Sci U S A. 2015; 112:3050–5. https://doi.org/10.1073/pnas.1500475112 PMID: 25713354
- Diallo M, Thonnon J, Fontenille D. Vertical transmission of the yellow fever virus by Aedes aegypti (Diptera, Culicidae): dynamics of infection in F1 adult progeny of orally infected females. Am J Trop Med Hyg. 2000; 62:151–6. https://doi.org/10.4269/ajtmh.2000.62.151 PMID: 10761742
- Burattini MN, Chen M, Chow A, Coutinho FA, Goh KT, Lopez LF, et al. Modelling the control strategies against dengue in Singapore. Epidemiol Infect. 2008; 136:309–19. <u>https://doi.org/10.1017/</u> S0950268807008667 PMID: 17540051
- Huang YJS, Higgs S, Horne KM, & Vanlandingham DL. Flavivirus-mosquito interactions. Viruses. 2014; 6(11): 4703–4730. https://doi.org/10.3390/v6114703 PMID: 25421894
- Gao D, Lou Y, He D, Porco TC, Kuang Y, Chowell G, et al. Prevention and Control of Zika as a Mosquito-Borne and Sexually Transmitted Disease: A Mathematical Modeling Analysis. Sci Rep. 2016; 6:28070. https://doi.org/10.1038/srep28070 PMID: 27312324
- He D, Gao D, Lou Y, Zhao S, Ruan S. A comparison study of Zika virus outbreaks in French Polynesia, Colombia and the State of Bahia in Brazil. Sci Rep. 2017; 7(1):273. <u>https://doi.org/10.1038/s41598-017-00253-1</u> PMID: 28325935
- Andraud M, Hens N, Marais C, Beutels P. Dynamic epidemiological models for dengue transmission: a systematic review of structural approaches. PLoS One. 2012; 7(11):e49085. https://doi.org/10.1371/ journal.pone.0049085 PMID: 23139836
- Chikaki E, Ishikawa H. A dengue transmission model in Thailand considering sequential infections with all four serotypes. J Infect Dev Ctries. 2009; 3:711–22. <u>https://doi.org/10.3855/jidc.616</u> PMID: 19858573
- Johansson MA, Arana-Vizcarrondo N, Biggerstaff BJ, Staples JE. Incubation periods of yellow fever virus. Am J Trop Med Hyg. 2010; 83(1):183–8. <u>https://doi.org/10.4269/ajtmh.2010.09-0782</u> PMID: 20595499
- Winnipeg Regional Health Authority (WRHA). Travel health and tropical medicines services. http://www.wrha.mb.ca/community/travel/; accessed on Mar, 2017.

- Massad E, Burattini MN, Coutinho FA, Lopez LF. Dengue and the risk of urban yellow fever reintroduction in Sao Paulo State, Brazil. Revista de Saude Publica. 2003; 37(4):477–84. https://doi.org/10.1590/ S0034-89102003000400013 PMID: 12937709
- He D, Dushoff J, Eftimie R, Earn DJ. Patterns of spread of influenza A in Canada. Proc Biol Sci. 2013; 280:20131174. https://doi.org/10.1098/rspb.2013.1174 PMID: 24026815
- Monath TP, Wilson DC, Lee VH, Stroh G, Kuteyi K, Smith EA. The 1970 yellow fever epidemic in Okwoga District, Benue Plateau State, Nigeria. I. Epidemiological observations. Bull World Health Organ. 1973; 49:113–21. PMID: 4545318
- **43.** Tomori O. Yellow fever: the recurring plague. Crit Rev Clin Lab Sci. 41:391–427. <u>https://doi.org/10.</u> 1080/10408360490497474 PMID: 15487593
- 44. World Health Organization. Fourth meeting of the Emergency Committee under the International Health Regulations (2005) regarding microcephaly, other neurological disorders and Zika virus. Saudi Med J. 2016; 37(10):1166–7.
- 45. Garske T, Van Kerkhove MD, Yactayo S, Ronveaux O, Lewis RF, Staples JE, et al. Yellow Fever in Africa: estimating the burden of disease and impact of mass vaccination from outbreak and serological data. PLoS Med. 2014; 11:e1001638. https://doi.org/10.1371/journal.pmed.1001638 PMID: 24800812
- van den Driessche P, Watmough J. Reproduction numbers and sub-threshold endemic equilibria for compartmental models of disease transmission. Math Biosci. 2002; 180:29–48. https://doi.org/10.1016/ S0025-5564(02)00108-6 PMID: 12387915
- Brauer F, Castillo-Chavez C, Mubayi A, Towers S. Some models for epidemics of vector-transmitted diseases. Infectious Disease Modelling. 2016; 1:79–87. https://doi.org/10.1016/j.idm.2016.08.001
- **48.** Keeling MJ & Rohani P. Modeling infectious diseases in humans and animals. Princeton University Press. 2008.
- 49. He D, Ionides EL, King AA. Plug-and-play inference for disease dynamics: measles in large and small populations as a case study. J R Soc Interface. 2010; 7:271–83. <u>https://doi.org/10.1098/rsif.2009.0151</u> PMID: 19535416
- King AA, Nguyen D, Ionides EL. Statistical inference for partially observed Markov processes via the R package pomp. J Stat Softw. 2016; 69:1–43. https://doi.org/10.18637/jss.v069.i12
- King AA. Statistical Inference for Partially-Observed Markov Processes. <u>http://kingaa.github.io/pomp/;</u> accessed on Mar, 2017.
- Schwarz G. Estimating the dimension of a model. Ann Stat. 1978; 6:461–4. https://doi.org/10.1214/aos/ 1176344136
- Allen LJ, Brauer F, Van den Driessche P, Wu J. Mathematical epidemiology. Lecture Notes in Mathematics. 2008; 1945:81–130. https://doi.org/10.1007/978-3-540-78911-6_3
- 54. Gao D, Amza A, Nassirou B, Kadri B, Sippl-Swezey N, Liu F, et al. Optimal seasonal timing of oral azithromycin for malaria. Am J Trop Med Hyg. 2014; 91:936–42. <u>https://doi.org/10.4269/ajtmh.13-0474</u> PMID: 25223942
- Lafferty KD. The ecology of climate change and infectious diseases. Ecology. 90:888–900. https://doi. org/10.1890/08-0079.1 PMID: 19449681
- 56. Reiter P. Climate change and mosquito-borne disease. Environ Health Perspect. 2001; 109 Supp 1:141–61. https://doi.org/10.2307/3434853 PMID: 11250812
- Fine PE, Clarkson JA. Measles in England and Wales–I: An analysis of factors underlying seasonal patterns. Int J Epidemiol. 1982; 11:5–14. https://doi.org/10.1093/ije/11.1.5 PMID: 7085179
- Bootsma MC, Ferguson NM. The effect of public health measures on the 1918 influenza pandemic in U. S. cities. Proc Natl Acad Sci U S A. 2007; 104:7588–93. https://doi.org/10.1073/pnas.0611071104 PMID: 17416677
- He D, Dushoff J, Day T, Ma J, Earn DJ. Inferring the causes of the three waves of the 1918 influenza pandemic in England and Wales. Proc Biol Sci. 2013; 280:20131345. <u>https://doi.org/10.1098/rspb.</u> 2013.1345 PMID: 23843396
- Svensson A. A note on generation times in epidemic models. Mathematical Biosciences. 2007; 208.1: 300–311. https://doi.org/10.1016/j.mbs.2006.10.010 PMID: 17174352
- del Carmen Marquetti Fernandez M, Flores YH, Nuviola DL. Spatial Distribution and Mainly Breeding Sites of Aedes aegypti (Diptera:Culicidae) in Luanda, Angola. Ann Community Med Pract. 2017; 3 (1):1017.
- Ribeiro H. Entomological studies during the 1971 yellow fever epidemic of Luanda, Angola. Mosquito News. 1973; 33(4):568–72.
- 63. Moreira R. Reining in Angola's yellow fever outbreak. 2016;716–717.

- **64.** President's Malaria Initiative Operational Plan, Angola. https://www.pmi.gov/docs/default-source/ default-document-library/malaria-operational-plans/fy17/fy-2017-angola-malaria-operational-plan.pdf? sfvrsn=7; accessed on Jul, 2017.
- Morrison AC, Minnick SL, Rocha C, Forshey BM, Stoddard ST, Getis A, Focks DA, Russell KL, Olson JG, Blair PJ, Watts DM. Epidemiology of dengue virus in lquitos, Peru 1999 to 2005: interepidemic and epidemic patterns of transmission. PLoS neglected tropical diseases. 2010; 4(5):e670. https://doi.org/ 10.1371/journal.pntd.0000670 PMID: 20454609
- World Health Organization. Yellow Fever Knowledge Pack. http://www.who.int/risk-communication/ yellow-fever/YF-Knowledge-pack.pdf?ua=1; accessed on Apr, 2017.